Author’s response to reviews

Title: Assessment of the feasibility and coverage of a modified universal hearing screening protocol for use with newborn babies of migrant workers in Beijing

Authors:

Beier Qi (beierqi@126.com)
Xiaohua Cheng (cxy010@163.com)
Hui En (enhui1971@163.com)
Bo Liu (trliubo@139.com)
Shichun Peng (pengshichun2000@sina.com)
Yong Zen (zhenyong914@sohu.com)
Zhenhua Cai (caizh258@yahoo.com.cn)
Lihui Huang (huangpub@126.com)
Luo Zhang (dr.luozhang@gmail.com)
Demin Han (handm@trhos.com)

Version: 7 Date: 20 July 2013

Author’s response to reviews:

A native-English speaker with scientific expertise has edited the language in our manuscript.